Author's response to reviews

Title: Standardization of surface electromyography utilized to evaluate patients with dysphagia

Authors:

Michael Vaiman (vaimed@yahoo.com)

Version: 3  Date: 29 May 2007

Author's response to reviews:

While the above mentioned correction have been already made, further corrections were as follows:
The full stop from the end of the title is removed.
Purpose was chnged to "Background" (in abstract)
References are cited in the text using consecutive numbers in square brackets starting from [1] etc.
I refer to all the figures within the text in a numerically consecutive order, starting from 1 onwards, but Fig. 4 is mentioned in text twice - before and after Fig. 5.
There is (and was) the acknowledgements section.
The Release letter (consent) of the model is attached.
References - "et al." is removed from the references and all authors are indicated in full.